Author’s response to reviews

Title: Etiologic spectrum and occurrence of coinfections in children hospitalized with community acquired pneumonia

Authors:

Wujun Jiang (eggiwjswjw@163.com)
Min Wu (18896788236@163.com)
Jing Zhou (zhoujsz@126.com)
Yuqing Wang (wang_yu_qingg@126.com)
Chuangli Hao (hcl_mdd@163.com)
Wei Ji (szdxjiwei@163.com)
Xinxin Zhang (zhangnewstar@126.com)
Wenjing Gu (guwenjing999@126.com)
Xuejun Shao (xuejunshao@hotmail.com)

Version: 2 Date: 28 Nov 2017

Author’s response to reviews:

Editor Comments:

1. We note that the current submission contains some textual overlap with other previously published works, in particular:

https://www.nature.com/articles/srep20246 which occurs in the in the ‘Subjects’ section of the Methods and https://www.nature.com/articles/srep39929 which occurs in the ‘Bacterial Detection’ section in the Methods.

While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work.
Please re-phrase these sections to minimise overlap.

response: we have done this.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

response: we have done this.